

# LJMU Research Online

Yang, Z, Lau, YY, Poo, MCP, Yin, J and Yang, Z

Port resilience to climate change in the Greater Bay Area

http://researchonline.ljmu.ac.uk/id/eprint/26126/

Article

**Citation** (please note it is advisable to refer to the publisher's version if you intend to cite from this work)

Yang, Z, Lau, YY, Poo, MCP, Yin, J and Yang, Z (2025) Port resilience to climate change in the Greater Bay Area. Transportation Research Part D: Transport and Environment, 142. ISSN 1361-9209

LJMU has developed LJMU Research Online for users to access the research output of the University more effectively. Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in LJMU Research Online to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.

The version presented here may differ from the published version or from the version of the record. Please see the repository URL above for details on accessing the published version and note that access may require a subscription.

For more information please contact <a href="mailto:researchonline@ljmu.ac.uk">researchonline@ljmu.ac.uk</a>

http://researchonline.ljmu.ac.uk/

Contents lists available at ScienceDirect

## Transportation Research Part D

journal homepage: www.elsevier.com/locate/trd

## Port resilience to climate change in the Greater Bay Area

Zhisen Yang<sup>a</sup>, Yui-yip Lau<sup>b</sup>, Mark Ching-Pong Poo<sup>c</sup>, Jingbo Yin<sup>d</sup>, Zaili Yang<sup>e,\*</sup>

<sup>a</sup> College of Urban Transportation and Logistics, Shenzhen Technology University, Shenzhen, China

<sup>b</sup> Division of Business and Hospitality Management, School of Professional Education and Executive Development, The Hong Kong Polytechnic

University, China

<sup>c</sup> Liverpool Hope Business School, Liverpool Hope University, Liverpool, UK

<sup>d</sup> Department of International Shipping, School of Naval Architecture, Ocean and Civil Engineering, Shanghai Jiao Tong University, Shanghai, China

<sup>e</sup> Liverpool Logistics, Offshore and Marine (LOOM) Research Institute, Liverpool John Moores University, Liverpool, UK

ARTICLE INFO

Keywords: Risk and resilience Climate change Objective Oriented Bayesian Network Port systems Port performance

#### ABSTRACT

As vital global supply chain nodes, ports are particularly vulnerable to risks caused by disruptions from climate-induced risks, e.g., extreme weather events, sea-level rise. Through incorporating the Objective Oriented Bayesian Network and Expectation Maximization approach, this paper develops a new framework which enables the assessment of port resilience to climate change across multiple dimensions. Taking the Greater Bay Area as an example, it is found that port systems are quite resilient under typical scenario, and the resilience could be enhanced with higher utilization rate and the implementation of recovery or adaptive measures with faster relative recovery speed. Key measures in improving the resilience of port systems from different perspectives are identified, including the advanced equipment, maintenance and reliability, and technology restoration in the future. Practically, such findings contribute to a deeper understanding of port resilience and offer industry practitioners and policymakers valuable implications to enhance sustainable and resilient port management.

#### 1. Introduction

The Greater Bay Area (GBA), encompassing the towns of Hong Kong, Shenzhen, and Guangzhou, has emerged as a thriving hub for container visitors and port improvement in Asia. As one of the four key bay areas worldwide,<sup>1</sup> this southern coastal region of China has become a worldwide logistics centre because of its strategic location, well-evolved infrastructure, and favourable rules (Xu et al., 2023). For example, the Port of Shenzhen handled around 28.5 million TEUs in 2023, making it an essential gateway for an alternative between China and the rest of the world (Port of Shenzhen, 2024), while the Port of Hong Kong has also maintained its significance with the management of around 16.5 million TEUs in 2023 (Port of Hong Kong, 2024). This exponential boom has been fuelled by the enlargement of economies, the upward thrust of e-commerce, and the place's integration into worldwide delivery chains. Ports in the GBA have become essential nodes, facilitating the seamless motion of products and helping the place achieve financial prosperity (Wang et al., 2022).

However, along with the economic prosperity brought by the coastal areas, it is found that container ports are usually affected by climate related events, e.g., growing temperatures, sea-degree upward thrust, natural disasters, resulting in potential risks and threats

\* Corresponding author.

E-mail address: z.yang@ljmu.ac.uk (Z. Yang).

<sup>1</sup> Four key bay areas refer to the San Francisco Bay Area, the New York Bay Area, the Greater Bay Area, and the Tokyo Bay Area.

https://doi.org/10.1016/j.trd.2025.104681

Received 26 November 2024; Received in revised form 27 January 2025; Accepted 23 February 2025

Available online 9 March 2025







<sup>1361-9209/© 2025</sup> The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

#### Z. Yang et al.

posed on critical port infrastructure systems (Wang et al., 2024a; Yang et al., 2018a). Such situation is particularly severe in the GBA, where the geographical location exposes it to many climate risks that have significant impacts on the regional development, evident by hundreds of published journal papers on climate risks in this region in recent years (e.g., Qin et al., 2024; Wang et al., 2024c; Zheng et al., 2024a).

Specifically, typhoons are the biggest climate risk that the ports in the GBA face, especially in the summer months. For example, in 2018, the hit of Typhoon Mangkhut caused a delay of over 300 container vessels in the Hongkong Port, resulting in huge economic loss (The Intergovernmental Panel on Climate Change, 2019). Statistically, in recent years, the annual number of typhoon warnings issued by Guangdong Province has witnessed an increase from around 10 in the past to 23 warnings in 2023 (Guangdong Meteorological Service, 2024). Such number is particularly higher than other key bay areas mainly due to the proximity to the tropical storm belt of the GBA. The rising frequency of typhoons and their increasing intensity are clear indicators of the growing climate risks faced by the coastal ports in the region. In addition, heavy rainfalls and storm surges are often connected with typhoons, which may put coastal ports at the risk of flooding, especially in low-lying port areas. As a result, coastal port systems will be largely affected, including the disruption of port logistics and the damage to port facilities. For instance, the storm surges and continuous heavy rainfall brought by Typhoon Mangkhut caused severe flooding in Guangdong Province, with the growth of water levels and the damage of port facilities in port areas (Sohu, 2018). Besides, other climate related risks, such as the rising sea levels and extreme temperatures, also affect the port management of the GBA to a certain extent in recent decades. Therefore, it is not difficult to find that ports in the GBA region are highly vulnerable to a wide range of climate related risks. Evaluating the port performance against these risks is essential for growing effective strategies and ensuring sustainable port resilience.

Within the context of port resilience, the Chinese government declared the "Dual Carbon Target" in 2020 and suggested an Action Plan for Carbon Dioxide Peaking before 2030 (Chen et al., 2023). Governments in the GBA region have responded to this call and implemented various policies aiming at mitigating the impacts of climate change and improving the port resilience performance, including the development of port adaptive strategies to withstand climate related risks in the formulation and amendment of Guangdong Province Climate Change Adaptation Action Plan, and the integration of climate change into the Port Master Planning. The implementation and continuous improvement of these standards is also a signal demonstrating the crucial importance of port resilience in achieving the sustainability goals locally and globally. Therefore, the GBA region is recognized as an emerging area where increasing climate risks (demand) meet the fast growth of climate adaptation policies (supply).

Driven by the demand for analyzing port resilience and the national strategies to promote sustainable port development, substantial research has been conducted with a particular consciousness on ports and weather extremes in the last decades (Wan et al., 2018; Wang et al., 2023b; Poo and Yang, 2024). However, significant research gaps have been found in current studies, i.e., the limitations of current applied approaches for the assessment of port network resilience to weather extremes, the demand for additional research studies on the resilience of ports in the GBA region where the effects of weather extremes can be extra severe, as well as the ignorance of experienced time of different port performance stages when facing climate related disruptions in the calculation of port resilience. More detailed information regarding these research gaps can be found in section 2.4.

To address these research gaps, this research aims to develop a comprehensive resilience assessment framework on port systems in the face of climate change using an Objective Oriented Bayesian Network (OOBN) model. Multidimensional factors that affect the resilience performance of port systems are considered, covering the disruptions, absorptive capacity, recovery capacity, and adaptive capacity. The questionnaire is designed to collect expert knowledge on the interdependencies among these factors, as well as to learn the parameters in the developed model. Taking the GBA as an example, the proposed OOBN model is proven to be effective in evaluating the preparedness of ports to withstand and recover from risks caused by disruptions dynamically, as well as calculating the exact port resilience value rationally. Sensitivity analysis is conducted to not only clarify the influence of different factors but more importantly to identify key factors that have significant influence on port resilience from different perspectives.

In this case, the research makes new contributions with regard to the following perspectives:

- The OOBN model is, for the first time, applied for the port resilience analysis upon the authors' best knowledge. It is proved to have better performance than the standard BN model in terms of the expansion of network structure, especially for port resilience assessment where various factors from multiple aspects are considered. Further, the application of the Expectation maximization (EM) learning approach provides a new solution to addressing the data uncertainty issue (i.e., missing data or abnormal data) in the calculation of conditional probabilities in the BN model existing in the port resilience field, which is superior to the state-of-the-art Noisy-or approach. Such incorporation can also reinforce the current knowledge on the port resilience field.
- 2) The newly proposed assessment framework in this research takes into account the experienced time of the disruption stage and the recovery stage when assessing port resilience, which was ignored in previous research. Such consideration could provide a more reliable assessment result because the time ratio between the reduction and recovery is undoubtedly a signal reflecting the resilience performance of a port system, highlighting another novelty of this research.
- 3) Clarification of the climate resilience variation of port systems in the GBA along with the change of key parameters. To enhance the port climate resilience, port governments are suggested to improve the utilization rate of port facilities and implement recovery strategies with faster relative recovery speed simultaneously.
- 4) Redefinition and of the newly introduced key items influencing the performance of port systems in terms of different dimensions of port resilience via a comprehensive analysis work incorporating an improved mutual information analysis approach and the scenario analysis. 'Reduced navigational safety in port areas' is identified as the risk factor that has the highest influence on the occurrence of climate-related disruptions, while key measures in improving the port climate resilience are identified from port

capacities of absorption, recovery and adaption, including the advanced equipment, maintenance and reliability, and technology restoration in the future respectively.

5) The research provides new valuable practical implications for strengthening and managing resilience in port management for port governments. For example, the formulation of specific instructions on controlling serious climate-related risks on port disruptions, the implementation of effective measures identified via the comprehensive analysis across both the recovery and adaption aspects.

The remainder of this paper is organized as follows. Section 2 reviews the relevant studies focusing on port risk and resilience. Section 3 introduces the research framework, and the major methodologies applied in this work. The case of the GBA is utilized in Section 4 to illustrate how the model is developed based on the collected data. Section 5 provides an in-depth analysis of the model outputs containing the assessment of port resilience and the sensitivity analysis of the involved factors. Section 6 concludes this work.

## 2. Literature review

In this section, current studies regarding the port resilience under climate change are discussed, as well as the application of Bayesian network (BN) for the quantification of the port resilience.

## 2.1. Port resilience

Critical infrastructure structures, consisting of container ports, are crucial for assisting societal features and allowing international exchange. These structures are characterized by their interdependence, complexity, and the cap potential for cascading failures (Rinaldi et al., 2001; Panahi et al., 2021). Container ports, as crucial infrastructure, play a pivotal function inside the international logistics network, facilitating the motion of products and assisting monetary improvement. However, the criticality of ports additionally puts them at risk of disruptions, such as the ones due to climate change. Understanding container ports' specific traits and demanding situations as crucial infrastructure sis important for growing powerful resilience strategies.

Traditionally, Port resilience is the performance of the port reacting and recovering from the risks associated with expected and unexpected disruptions, or in other words, the ability to prevent the exposure of the disruptions (Lau et al., 2024). It consists of four dimensions: disruption, absorptive capacity, recovery capacity, and adaptive capacity (Li et al., 2024a; Wang et al., 2023a).

Disruptions refer to those events that may cause negative impacts on the port, e.g., substantial financial losses, or severe imbalances of the port management system (Gu & Liu, 2023). These disruptive events could be viewed as associated risks threatening various aspects of the normal operations of coastal ports. The cap potential disruptions that ports may also face because of climate change are manifold and may have far-accomplishing consequences (Liu et al., 2023b). Sea-stage upward thrust, for example, can cause the inundation of port facilities, rendering them unusable and disrupting maritime supply chains (Becker et al., 2013). Extreme climate activities, consisting of hurricanes, typhoons, and hurricane surges, can cause extreme harm to port infrastructure, such as breakwaters, quay walls, and load coping with equipment. These disruptions to port infrastructure will have cascading consequences on international maritime supply chains, as ports function as crucial nodes inside the motion of products (Zhao et al., 2024). The failure of an unmarried port can cause delays, bottlenecks, and the disruptions that ports face is crucial for growing powerful resilience strategies (Liu et al., 2018). Understanding the precise climate change-brought disruptions that ports face is crucial for growing powerful resilience strategies (Liu et al., 2023c). Besides, enhancing the resilience of ports in the face of climate change is important for ensuring the continuity of world exchange and monetary prosperity (Wan et al., 2024).

When disruptive events occur, the port performance will be affected. At this time, the intrinsic ability of the port will help absorb the negative impacts caused by these disruptions, which is called the absorptive capacity (Golan et al., 2020). This consists of measures consisting of infrastructure reinforcement, flood safety structures, and emergency reaction planning. For example, the Port of Rotterdam in the Netherlands has carried out a complete flood safety system, such as hurricane surge boundaries and dike reinforcement, to decorate its absorptive capability and face up to the effects of sea-stage upward thrust and excessive climate activities (Port of Rotterdam, 2022). Similarly, the Port of Los Angeles has invested in the improvement of shore electricity infrastructure, permitting ships to connect with the electric grid and decrease their emissions even at berth, thereby mitigating the environmental effect of port operations (Port of Los Angeles, 2021).

Subsequently, the port performance will recover from its lowest point gradually and steadily because of its recovery capacity (Wang et al., 2024b). The idea of recuperation potential and its importance in making sure to spark off recovery of port capabilities after disruptions must be discussed. The operation generally takes longer than what it undergoes in absorption. It may return back to its original condition, enhance to a higher condition, or attain a lower condition (Rehak et al., 2018; Gu et al., 2024). Examples of successful recuperation plans and techniques hired via way of means of resilient ports can offer precious insights. The Port of Singapore, for instance, has advanced a complete enterprise continuity plan that outlines tactics for the speedy recovery of port operations on the occasion of a disruption (Maritime and Port Authority of Singapore, 2022a; Maritime and Port Authority of Singapore, 2022b). This consists of the pre-positioning of crucial gadgets and the status quo of backup centres to ensure the continuity of shipment management and logistics services.

The port system will then maintain at the evolved steady level, and adapt to the disruptions it has experienced spontaneously, indicating it will learn from such progress and react or recover faster and more effectively next time in case similar disruptions happen (Xu et al., 2024). Case research highlighting powerful adaptive measures carried out via means of ports globally can indicate the improvement of complete resilience evaluation frameworks (Xia et al., 2024). The Port of Shenzhen in China, for example, has followed a proactive technique to weather trade adaptation, together with the implementation of inexperienced infrastructure, the

diversification of strength sources, and the mixing of climate change issues into its long-term period planning (Port of Shenzhen, 2024). These measures have more advantageous the port's cap potential to develop to the evolving weather panorama and reduce the effect of destiny disruptions.

## 2.2. Assessment methods for port resilience

Recent years, a wide variety of approaches has been applied to assess port resilience, e.g., decision-making approaches (John et al., 2014), simulation analysis (Zhou et al., 2021), and other quantitative approaches. Among these methods, BN is a preferred one for the assessment of port resilience because of its ability in dealing with causal relationships and risk assessment (Yang et al., 2024a).

Incorporating the Fuzzy Analytical Hierarchy Process with the BN, John et al. (2016) presented a quantitative risk assessment model to assist port governments in implementing strategies for improving the resilience of port systems. Focusing on inland waterway ports, Hosseini & Barker (2016) proposed a useful tool based on the BN model in quantifying port resilience. Later, Hossain et al. (2019; Hossain et al., (2019, 2020) developed two BN models to assess the resilience of full-service deep-water ports and surrounding supply chain network of inland ports respectively. Similarly, Panahi et al. (2022) identified the key disruptions and strategies affecting the container terminals in Hongkong port. Wang et al. (2023a) constructed a port resilience assessment model, and they compared the performance of automated and non-automated terminals in Shanghai Yangshan port in China.

## 2.3. Adaptation and mitigation strategies in climate risk management for ports

Seaports play a critical role in global trade and are increasingly vulnerable to the impacts of climate change, including extreme weather events, sea-level rise, and natural disasters. To address these challenges, the literature highlights the need for adaptation and mitigation strategies to enhance seaport operations' resilience and sustainability (Morris, 2020). This review synthesises existing research on adaptation and mitigation (decarbonisation) in disaster management and climate risk analysis for seaports, emphasising both theoretical and practical contributions.

Adaptation strategies focus on increasing the resilience of seaports to climate-induced disruptions. Ng et al. (2018) highlight the importance of understanding sector-specific climate impacts to develop appropriate adaptation measures for commercial ports. They emphasise stakeholder collaboration, policy integration, and the alignment of adaptation goals with long-term business strategies. Jiang et al. (2020) distinguish between mitigation and adaptation strategies for seaports, underscoring the need to balance short-term disaster responses with long-term infrastructure resilience. Gong et al. (2020) extend this perspective by examining seaport investments in capacity expansion and natural disaster prevention, revealing that proactive investments can significantly reduce vulnerability to disasters while maintaining operational efficiency. Griese et al. (2021) construct a theoretical framework to analyze the acceptance phases for climate adaptation measures and disclose the key points for promoting the utilization of different adaptation measures. Wang et al. (2024b) develop an evaluation model to analyse the adaptation measures of ports in respond to typhoons, and reveal the inadequate supply of institutions and the absence of the main port stakeholders have higher probabilities leading to imperfections of port systems. Wu et al. (2024) explore how co-opetition with dry ports can enhance seaports' disaster preparedness, resource sharing, and operational continuity during disruptions. Roukounis and Tsihrintzis (2024) also explore the effectiveness of different adaptation measures in enhancing port resilience.

Recent studies have also introduced the Climate Change Risk Indicator (CCRI) framework as a comprehensive tool for assessing climate risks in transport systems. Poo et al. (2021) focused on the application of CCRI for UK seaports, providing insights into the prioritisation of resilience measures and enabling more targeted adaptation strategies for port operations. Similarly, Wang et al. (2023b) apply the CCRI framework to a multi-modal transport system, highlighting its effectiveness in identifying and mitigating vulnerabilities posed by climate change. These frameworks offer valuable methodologies for integrating risk assessment into seaport disaster management and climate adaptation planning.

Mitigation strategies in seaport operations primarily focus on reducing greenhouse gas emissions and promoting sustainability. Masodzadeh et al. (2024) analyse the contribution of ports to shipping decarbonisation through incentive programs and the role of port state control. Their research demonstrates the potential of policy-driven initiatives to accelerate the adoption of low-carbon technologies and practices within the maritime sector. Alamoush et al. (2024) investigate the determinants of port decarbonisation implementation, using Implementation Theory to identify key enablers and barriers. Their study highlights the significance of regulatory frameworks, stakeholder engagement, and financial incentives in achieving decarbonisation goals. Similarly, Calderón-Rivera et al. (2024) address barriers to sustainable inland waterway transport, offering solutions that can inform broader decarbonisation efforts in the maritime industry.

A comprehensive approach to climate risk analysis often requires the integrity of both adaptation and mitigation strategies. Zheng et al. (2024) advocate implementing demonstrator projects to showcase effective adaptation measures for transport facilities, including seaports. Their research emphasises the importance of pilot initiatives in building stakeholder confidence and guiding large-scale implementation. Wu et al. (2024) highlight the interplay between adaptation and mitigation through collaborative initiatives with dry ports, suggesting that co-opetition frameworks can simultaneously enhance disaster resilience and reduce carbon footprints. This dual focus aligns with the broader trend of integrating climate adaptation and mitigation into seaport risk management strategies.

#### 2.4 Research gaps

The relevant studies mentioned above contributed a lot to analysing port resilience from multiple dimensions. However, notable research gaps have emerged in the following aspects:

1) The limitations of current applied methods for port resilience assessment.

As indicated by previous research in this field, the standard BN model is the most popular method for the assessment of port resilience because of its capacities in dealing with risks and uncertainties. However, since the port resilience assessment is a complicated work involving various factors from multiple aspects, it will be intractable for the standard BN model to present a reliable network with expanded network structure (Fu et al., 2023). Therefore, new method is required to address this technical issue.

2) The demand for port resilience assessment in the GBA.

As stated in the introduction section, the port areas in the GBA region are largely affected by the climate risks due to its geographical location, hence making it necessary and significant to develop a systematic framework for quantifying the port resilience performance and analyzing the effect of risks on port systems brought by climate change. However, such topic is rarely discussed in previous works, making it a gap to fulfil in academic field.

3) The ignorance of experienced time of different stages.

Another research gap has emerged in the context of undertaking influenced time of different stages when calculating port resilience. Longer experience time in disruption and absorption stages or shorter experience time in recovery and adaption stages means better resilience performances of port systems. However, previous works ignored this point, which may affect the reliability of the assessment results.

Addressing these gaps could help enhance the understanding of port resilience against climate change for both the industrial and academic fields.

## 3. Methodology

This paper proposes a new idea of assessing resilience-constructing capability through three key dimensions: absorptive capability, recovery capability, and adaptive capability. The formulation of the idea is also supported by the real-world observations and previous studies against each dimension in a piecemeal manner. It is our intention to combine them in one holistic framework for comprehensive assessment of port climate resilience.

#### 3.1. Quantification of port resilience

Through analysing the progress of port performance when disruptions occur, it is obvious that the whole process can be divided into two stages. In the first stage, because of the risks brought by disruptions and the absorptive capacity of the port system, port performance may experience a certain degree of loss; and in the second stage, the recovery capacity and the adaptive capacity will improve the port performance and reinvent the port system simultaneously. As a result, port resilience is commonly recognized as the ratio of the recovered performance (RP) to the lost performance (LP), as suggested by relevant studies in this field (Henry & Ramirez-Marquez, 2012; Hosseini & Barker, 2016; Panahi et al., 2022). In fact, since both the LP and the RP will occur only when certain conditions are satisfied, it is more appropriate to calculate the port resilience using the expected values of the LP and the RP considering the occurrence probabilities of corresponding conditions.

That is to say, let the port performance before the occurrence of disruption be x1, the performance after the disruption be x2, the occurrence probability of disruption be P1, the probability of absorptive capacity taking effect be P2. It is obvious that the port will lose performance (x1 > x2) only when the disruption occurs and the absorptive capacity of the port is unable to absorb such impacts, otherwise the lost performance would be zero as the disruptions are absorbed by the port (x1 = x2). Hence, the expected LP is calculated as:

$$LP = P_1(1 - P_2)(x_1 - x_2)$$
(1)

Further, during the port performance moves from  $x_1$  to  $x_2$ , the recovery and adaptation abilities of the port system will consistently prompt the port performance to recover back to a normal operational status. Such recovery progress will occur only when either one of the recovery/adaptation abilities performs successfully, and the recover degree is set as a, which is the utilization rate of the port under regular operation situations. Besides, let the success rate of recovery capacity and adaptive capacity be  $P_{3}$ , and  $P_4$  respectively, then the expected RP is calculated as:

$$RP = P_1(1 - P_2)\{[1 - (1 - P_3)(1 - P_4)] \times \alpha(\mathbf{x}_1 - \mathbf{x}_2)\} = [1 - (1 - P_3)(1 - P_4)] \times \alpha LP$$
(2)

As mentioned above, the traditional way to obtain the port resilience is to calculate the ratio of RP to LP subsequently. However, such classical expression utilized in previous research ignores one important element when assessing resilience, which is the influenced time at different stages. Under the condition of same disruption, a port system whose performance reduces slower in the first stage and rebounds or recovers faster in the second stage undoubtedly has stronger resilience than those systems with opposite trends. Therefore, it is necessary to take the influenced time at different stages into consideration when assessing port resilience.

In this research, let the performance reduction time be  $t_1$ , the recovery time till the steady status is  $t_2$ , then the port resilience is assessed by Equation (3), where it is proportional to  $t_1$  and inversely proportional to  $t_2$ . The ratio  $t_1/t_2$  is named as relative recovery

speed (rrs), which represents the recovery speed comparing with the disruption speed.

$$Port Resilience = \frac{RP}{LP} \times \frac{t_1}{t_2}$$

$$Port Resilience = [1 - (1 - P_3)(1 - P_4)] \times \alpha \times \frac{t_1}{t_2}$$
(3)

## 3.2 Research framework

To achieve the assessment of port resilience, a research framework based on the novel application of the OOBN is proposed in Fig. 1. An incorporation of the Expectation Maximization (EM) approach and improved mutual information approach is utilized, and a



Fig. 1. Research framework for the port resilience assessment facing climate change.

detailed explanation of the framework is presented in the ensuing section.

#### 3.2.1. Step 1: Variable identification and data acquisition

In this research, since the objective is to assess port performance in terms of the disruptions caused by climate change, relevant events and measures that may affect port resilience in this aspect are identified as the variables in the OOBN model from the literature. A questionnaire is then designed focusing on the causal relationships and dependencies among these variables corresponding to different aspects of port resilience. Various experts in this field are invited to express their opinions, which could help develop the subsequent model.

Since different experts have different research backgrounds, e.g., educational level, working experience, different weights should be assigned to these experts. In this research, the weight of each interviewed expert is calculated based on the following equation:

$$w_i = \frac{s_i}{\sum_{i=1}^n s_i}, i = 1, 2, ..., n$$
(4)

Where *n* is the number of all involved experts,  $w_i$  is the weight for the selected expert,  $s_i$  is the weight score of the selected expert calculated based on the criteria presented in Table 1, which is set according to many similar studies (Senol et al., 2015; Kuzu et al., 2019; Gürgen et al., 2023), and  $\sum_{i=1}^{n} s_i$  is the total weight scores of all experts. In a word, the weight of each expert is calculated as the proportion of total weight scores. The determination of expert weights is able to help calculate the prior probabilities of root variables.

## 3.2.2. Step 2: Development of the OOBN model

As listed in literature review section, there have been several studies assessing port resilience using the standard BN in recent years. Because of its advantages in modelling risk and uncertainties through a probabilistic graphic network, the application of the BN is a proper way to assess and analyze port risk and resilience from both forward risk assessment and backward diagnosis (Yang et al., 2018b; Yang et al., 2023a).

However, because port resilience is a complicated problem involving influential factors from multiple aspects, i.e., risks and disruption, absorptive capacity, adaptive capacity, and recovery capacity, it is evident that the standard BN models are unable to manage the associated variables in a single network simultaneously in terms of the expansion of network structure (i.e., more involved factors) in this research.

To solve such an issue, an extended version of BN, called OOBN, is, for the first time, applied to port resilience analysis upon the authors' best knowledge. It is a complicated model consisting of several sub-models (Liu et al., 2016; Fu et al., 2023). Fig. 2 illustrates an example of an OOBN model. Each sub-model is a standard BN model, presenting the causal relationships among variables under this aspect. In each sub-model, there are three types of variables: input variable, internal variable, and output variable. For input variables and output variables, they work as the interfaces of each sub-network, or in other words, the input variables of the subsequent sub-network are linked with the output variables of the former sub-network, thus connecting the whole network. While for the internal variables, they are invisible and not allowed to connect with variables in other sub-networks. Additionally, the basic principle of the standard BN still applies to OOBN, which means loops are not allowed either in a single sub-network or the whole network.

Because of the distinctive features of OOBNs, it can achieve object-oriented modeling (Fu et al., 2023). Each sub-network is

Weight	scores	for	experts.

Table 1

Constitution	Classification	Score
Current Job Position	Others	1
	Director/Analyst	2
	Manager	3
	Consultant	4
	Supervisor/Officer	5
	Others	1
Educational level	Bachelor	2
	Master	3
	Doctoral	4
Working Department	Others	1
	Research and Development	2
	Transportation	3
Company's Nature	Others	1
	Third-party logistics service providers/Freight forwarder	2
	Transport operators	3
	Port	4
	2–5 years	1
Working experience	6–9 years	2
	10–15 years	3
	16–20 years	4
	21–25 years	5
	26–30 years	6
	31 years or above	7



Fig. 2. An example of an OOBN model.

abstracted using an instance node, and the information of one sub-network is transmitted forward to the next one via the connections between input and output variables. Moreover, it also possesses the basic functions of the standard BN such as forward assessment and backward diagnosis. All of these natures make the OOBN a suitable modelling approach to the assessment of port resilience in this research.

To be specific, the development of the OOBN model in this research consists of two parts: structure determination and parameter learning.

## (1) Structure determination

The determination of model structure is easy to achieve. Based on the identified variables in Step 1, the structure of each subnetwork is obtained according to the discussions and opinions from invited experts, resulting in four sub-networks corresponding to the four dimensions of port resilience. Then the four target variables in four sub-networks are combined and act as the input variables for the final sub-network whose target variable is 'port resilience', as presented simply in Fig. 1.

(2) Parameter learning.

Table 2

Once the structure of the OOBN model is determined, parameter learning is conducted to clarify the probability tables for each variable embedded in the model. There are two types of probability tables in the model, one is the prior probability tables (PPTs) for root variables (variables with no parent nodes), and the other is the conditional probability tables (CPTs) for target variables or internal variables with parent nodes. PPTs are relatively easy to acquire. Since each interviewee is asked to present their opinion on the occurrence probability of these root variables, hence the prior probabilities are calculated as the weighted average value of their answers, where the weights are determined based on the different backgrounds, experiences, and professionalism of the interviewees.

For the determination of CPTs, the traditional way to calculate conditional probabilities in previous research in the port resilience

Parameters in the EM algorithm.			
Parameters	Descriptions		
X <sub>i</sub>	Variables		
x <sub>ik</sub>	$k^{th}$ state of $X_i$		
Parents(X <sub>i</sub> )	parent nodes of $X_i$		
$Parents(X_{ij})$	$j^{th}$ state of $Parents(X_i)$		
N <sub>ijk</sub>	Number of cases involving $x_{ik}$ and $Parents(X_{ij})$ in training data		
$c_n$	Specific case of $x_{ik}$ and $Parents(X_{ij})$ in the training data, value range: $[1, N_{ijk}]$		
α	Conditional probability of $x_{ik}$ under the condition of $Parents(X_{ij})$		
â	Estimation of $\alpha$		
â	Updated value of $\hat{\alpha}$		
d	Data sample		
$d^*$	Missing data sample		
E	Desired degree of precision		

field is the Noisy-or approach (Hosseini & Barker, 2016). However, since there exists missing data or abnormal data if and when some interviewees may be unfamiliar with some aspects, the Noisy-or approach is incapable of modelling the associated uncertainty in data and hence not selected in this research. Instead, the EM learning approach, which is a deterministic approach to learning parameters asymptotically, is applied to calculate the conditional probabilities in the proposed model. It is again for the first time applied in the port resilience field, upon the authors' best knowledge, providing a new solution to the difficulty of accommodating high uncertainty in data in BN CPT of port resilience beyond the state-of-the-art methods (i.e., Noisy-or). Based on the probability distributions given by interviewees in the collected questionnaire data, a learning database containing simulated cases conforming to these probability distributions is generated accordingly, which could be utilized to achieve the CPT learning work via the application of the EM learning approach. It has been proven to be effective in dealing with incomplete data issues (Yang et al., 2024a; Yang et al., 2024b). The detailed information for the approach is introduced as follows.

The EM learning approach is briefly introduced as follows (Lauritzen, 1995). Table 2 presents the defined parameters utilized in the approach, while the subsequent procedures illustrate how the EM learning approach is achieved. More detailed information is documented in Yang et al. (2023b), including the detailed steps on how it is conducted for generating CPTs.

- 1) Set the value of  $\hat{\alpha}$  and  $\in$  arbitrarily.
- 2) The expected sufficient statistic for  $d^*$  is calculated through Equation (2)
- 3) The updated value  $\hat{\alpha}'$  for  $\alpha$  is obtained following the rule of maximising  $P(d^*|\hat{\alpha}')$  using the calculated  $E_{P(X_i|d,\hat{\alpha})}N_{ijk}$ .

$$E_{P(X_i|d,\hat{a})}N_{ijk} = \sum_{n=1}^{N_{ijk}} P(\mathbf{x}_{ik}, Parents(X_{ij})|c_n, \widehat{\alpha})$$
(5)

$$\widehat{\boldsymbol{\alpha}'}_{ijk} = \frac{E_{P(X_i|d,\hat{\boldsymbol{\alpha}})} N_{ijk}}{\sum_{k'} E_{P(X_i|d,\hat{\boldsymbol{\alpha}})} N_{ijk'}}$$
(6)

 $\operatorname{Set}\widehat{\alpha'} = \widehat{\alpha'}_{iik}.$ 

4) When  $|\widehat{\alpha'} - \widehat{\alpha}| > \epsilon$ , replace  $\widehat{\alpha}$  with  $\widehat{\alpha'}$ .

5) Iterate this procedure until  $|\hat{\alpha} - \hat{\alpha}| \leq \epsilon$ . The value of  $\hat{\alpha}$  at this time is the learned value for the parameter.

Consequently, the calculated CPTs together with the obtained network structure constitute the final OOBN model.

## 3.2.3. Step 3: Analysis

Comprehensive analysis is conducted after the model is developed, which consists of the scenario simulation for the assessment of port resilience performance under various conditions, and sensitivity analysis for the identification of key variables that affect port performance in different aspects of resilience the most.

The scenario simulation is simply accomplished via inputting the relevant information into the proposed model, resulting in the corresponding value of port resilience under specific scenarios, while the sensitivity analysis is achieved based on the application of an improved mutual information analysis approach which is introduced as follows.

In information theory, mutual information is defined as the information shared by a pair of variables, which is an effective way to measure the strength of relationships between different variables (Fan and Yang, 2024; Li et al., 2024b; Zhou et al., 2024). A higher value of mutual information is usually recognized as a signal of a strong relationship between the selected variables, thus help determining the key variables with significant influence on target variable.

However, considering the entropy of two variables is different, it would be one-sided to determine the strength of relationship based on the mutual information value if the difference of entropy value between X and Y is huge. Hence, in order to eliminate the entropy value differences of different variables, an improved mutual information (IMI) value is introduced in lEquation 7, which is able to clarify the relationships between attribute variables and target variable reliably and comprehensively.

$$IMI = \frac{I(X,Y)}{\sqrt{H(X) \times H(Y)}}$$
(7)

Where I(X, Y) is the mutual information between variable X and Y, H(X) and H(Y) are the entropy of variable X and Y. After the calculation of the IMI value between each attribute variable and the target variable under the condition of the given data, variables with higher values are considered to have a greater influence on certain perspectives of the port resilience.

## 4. Model development - A case in the greater Bay Area

## 4.1. Data

A questionnaire is designed to gather relevant data used for the model development according to the approaches proposed by Iacobucci and Churchill (2010), which consists of seven steps: 1) Selecting the required necessary information from relevant literature; 2) Adopting a relevant question categorization and administrative manner; 3) Employing a relevant question layout and components; 4) Correctly organizing the question response; 5) Deploying precise wording in each question; 6) Selecting a practical and structured

series of questions; 7) Generating a decision on the relevant design of questionnaire survey. After the adjustment and pre-testing work, a comprehensive layout of the questionnaire will be developed eventually.

The survey questions were discussed with relevant stakeholders including researchers, environmentalists, logistics associations, and industrial practitioners (e.g., logistics firms, terminal operators, and transport operators) to identify proper content, quantify the model, and question design. In doing so, it can increase the validity of the content and improve the correctness of the survey instruments. In particular, fuzzy wordings and duplex questions have been fully eliminated. A group of target survey respondents are also invited to conduct a pilot test of the survey. This step is commonly described as face validity (Ngai et al., 2008).

In general, the survey questionnaire is split into four key parts:

*Part A*: The profile of survey respondents is collected to give demographic or background information including gender, age, education level, current job position, the working department, the company's nature, and working experience in the logistics industry. The survey respondents were required to answer the seven items with close-ended questions.

*Part B*: In this part, major port disruptions associated with climate-related risks are identified. As described in the introduction section, port areas in the GBA are vulnerable to climate-related risks because of its geographical location. Among all possible risks that port areas may face, some of them are acute ones caused by extreme weather events such as the heavy rainfalls, storm surges, flooding, while others are chronic productions of long-term climate pattern changes like sea-level rise and temperature increase. Through a careful investigation on relevant pervious research focusing on these involved risks in two different types (e.g., Hosseini and Barker, 2016; Becker et al., 2018; Patel et al., 2018; Hossain et al., 2019; Hosseini et al., 2019; Verschuur et al., 2020; Izaguirre et al., 2021; Notteboom et al., 2021; Poo et al., 2021), nine major disruptions of port infrastructure were identified and verified to be relevant to the context of the GBA, including the disruption of maritime supply chains, damage to nearby ecosystems, damage to port infrastructure, increased maintenance costs, increased downtime and delays, increased risk of ship collision in port areas, reduced navigational safety in port areas, reduced cargo handling capacity, and power outages.

As such, survey respondents were asked to assign a probability of occurrence for each disruption based on their perception and/or experience about container ports in the GBA. Accordingly, the survey respondents were asked to indicate a probability of occurrence for every disruption ranging from 0 to 20 % – very low', 21 % to 40 % – 'low', 41 % to 60 % – 'medium', 61 % to 80 % – 'high', and 81 % to 100 % – 'very high'.

As such, survey respondents were asked to assign a probability of occurrence for each disruption based on their perception or experience about container ports in the GBA. Accordingly, the survey respondents were asked to indicate a probability of occurrence for every disruption ranging from 0 to 20 % – very low', 21 % to 40 % – 'low', 41 % to 60 % – 'medium', 61 % to 80 % – 'high', and 81 % to 100 % – 'very high'.

*Part C*: In this part, the resilience-building capacities of ports in the GBA in response to the climate-induced risks are assessed. Aiming at reducing the vulnerability and enhancing preparedness of port systems, different measures are proposed from multiple dimensions, i.e., the absorption, recovery, and the adaptation. According to the guidelines for resilient port operations and adaptations against climate change formulated by various international organizations, e.g., International Maritime Organisation (IMO), World Meteorological Organization (WMO), United Nations Conference on Trade and Development (UNCTAD), International Labor Organization (ILO), these measures could be classified into several categories: technological measures, social and community measures, policy and planning measures, and operational measures. Focusing on these categories, associated measures from three dimensions are recognized from related previous research (e.g., Mansouri et al., 2010; Madhusudan and Ganapathy, 2011; Hosseini et al., 2019; Komugabe-Dixson et al., 2019; Panahi et al., 2021), resulting in five measures in the absorptive capacity, five measures in the recovery capacity, and seven measures from the adaptive capacity presented in the following Table 3a and b.

In response, survey respondents were asked to assign a probability of occurrence for each capacity based on their perception and/or experience about container ports in the GBA. Similar to Part B, survey respondents were asked to give a probability of occurrence for each measure ranging from 0 to 20 % – 'very low', 21 % to 40 % – low', 41 % to 60 % – 'medium', 61 % to 80 % – 'high', and 81 % to 100 % – 'very high'.

*Part D*: At last, related experts are invited to discuss the causal relationships among the identified factors from different perspectives. The pairs of factors that are supposed to have relationships will be linked in the model, constituting the structure of subnetworks and the finalized model.

Once the questionnaire is designed appropriately, a purposive sampling approach is selected to choose the proper population members for this research. The survey is distributed to target respondents online from December 2023 to January 2024, resulting in 100 valid questionnaires at last. The survey questionnaires were completed anonymously, and the responses were kept confidential and used only for academic purposes. The profile of respondents is summarized and presented in Appendix A. Specifically, 64 % of respondents were male and 36 % of respondents were aged between 26 and 35 years old. The majority of respondents (93 %) were well educated and acquired at a tertiary educational level (i.e., at least Associate Degree or Higher Diploma level). Moreover, 40 % of

#### Table 3a

Measures for improving resilience-building capacities.

Absorptive	Advanced equipment, Electronic exchange platform, Skillful emergency response team, Skilled labor with professional qualification,
measures	Communication
Recovery	Maintenance and reliability, Port innovation, Port health and safety management, Business continuity plan, Crisis management policy
measures	
Adaptive measures	Technology restoration, Port worldwide network, Port community system, Hinterland connection, Service restoration, Data
	interoperability standardization, Cyber port infrastructure

Table 3b	
IMI value	of influencing variables

Perspective	Variable	IMI
Disruption	Reduced navigational safety in port areas	0.02362
	Disruption of maritime supply chains	0.01632
	Damage to nearby ecosystems	0.01271
	Increased maintenance costs	0.01155
	Increased downtime and delays	0.01067
	Reduced cargo handling capacity	0.00844
	Damage to port infrastructure	2.80*e <sup>-10</sup>
	Increased risk of ship collision in port areas	3.51*e <sup>-11</sup>
	Power outages	4.55*e <sup>-13</sup>
Absorptive capacity	Advanced equipment	0.04533
	Electronic exchange platform	0.04403
	Skillful emergency response team	0.03761
	Skilled labor with professional qualification	0.01147
	Communication	0.01023
Recovery capacity	Maintenance and reliability	0.08039
	Port innovation	0.07652
	Port health and safety management	0.05885
	Business continuity plan	8.21*e <sup>-11</sup>
	Crisis management policy	5.03*e <sup>-11</sup>
Adaptive capacity	Technology restoration	0.06447
	Port worldwide network	0.04596
	Port community system	0.01723
	Hinterland connection	0.01677
	Service restoration	0.00988
	Data interoperability standardization	2.88*e-10
	Cyber port infrastructure	2.64*e-10

respondents were positioned at a managerial level. In addition, 37 % of respondents indicated that their working department is the transportation department, while 28 % of respondents expressed that their company nature is third-party logistics service providers. Furthermore, 34 % of respondents pointed out that they have worked for 2–5 years in the logistics industry.

## 4.2. Development of OOBN model

## 4.2.1. Development of sub-networks

The OOBN model consists of five sub-networks:  $S_1$  (target variable: disruption),  $S_2$  (target variable: absorptive capacity),  $S_3$  (target variable: recovery capacity),  $S_4$  (target variable: absorptive capacity), and  $S_5$  (target variable: port resilience). As mentioned in section 3, the relationships among involving variables in each sub-network are figured out via the expert feedbacks.

Once the structure is determined, parameter learning is conducted subsequently. The prior probabilities of variables are calculated



Fig. 3. Sub-network of 'Disruption'.

as the weighted average of the interview outputs based on the weight calculation in Equation (4). As for the CPTs of other variables, they are acquired through the EM learning approach. Based on the calculated prior probabilities of each influencing variable, a data simulation process (Jensen and Nielsen, 2007) is conducted in the *Hugin* software to generate a learning database conforming to these numbers. EM learning approach is then applied to achieve the learning of CPTs existing in these sub-networks.

Consequently, the four sub-networks are presented in Figs. 3 to 6.

### 4.2.2. Development of finalized OOBN model

After the four sub-networks are determined, the finalized OOBN model is developed subsequently. Each target variable corresponding to each dimension in the above-developed four sub-networks is used as the input variable of the last sub-network, whose target variable is the 'port resilience'.

The last sub-network has three layers: four dimensions (i.e., disruption, absorptive capacity, recovery capacity, and adaptive capacity) of port resilience are located at the input layer, the LP and RP are set as the utility node in the internal layer, and the target variable 'Port resilience' is set as a function node putting at the last layer alone. The calculation of LP, RP, and Port resilience is achieved based on Equations (1) to (3) in Section 3.1. To be specific, the last sub-network is presented as follows in Fig. 7.

Consequently, by combining all sub-networks together, the finalized OOBN model for the assessment of port resilience in the GBA area is illustrated in Fig. 8.

Where each instance node (rounded rectangles in the figure) in the model represents the corresponding sub-network that is put forward in Sections 4.2.1 and 4.2.2, whose name is presented at the top of the rectangle. Meanwhile, the links in the model refer to the connections between the output variables and input variables of different sub-networks, conforming to the features of the OOBN model.

## 5. Results and analysis

## 5.1. Model results analysis

Through a thorough investigation of the proposed OOBN model, it is found that there is a 49.03 % chance that the disruption would occur in the port cluster in the GBA region, if and when a solid climate risk event occurs. Among all possible disruptions, the increased maintenance costs are identified as the risk factor that is most likely to happen (54.88 %), while power outages are the one that is most unlikely to happen (35.49 %). The occurrence probabilities of other disruption types fall within the range of 40 % to 50 %.

The structure of the absorptive capacity sub-network is the simplest one, with all influencing variables acting as independent nodes pointing at the absorptive capacity of the port. The port system in the GBA has a 69.37 % chance of success in absorbing the shocks and negative impacts caused by the disruptions, as indicated by the model results. All absorptive measures are recognized to have higher occurrence probabilities over 60 %. Of course, this result is based on the marginal probability, concerning the likelihood of all concern climate risk events. If and when a specific climate risk event is identified, such a result will be updated to better reflect specific cases (see Section 5.2).

For recovery capacity and adaptive capacity, both capacities have a success chance of around 70 % to achieve the recovery and adaptation of port performance. Considering the potential measures for these two perspectives, measures proposed for the adaptive capacity normally have higher occurrence probabilities, while those for the recovery capacity have a relatively lower chance. Data interoperability/standardization is viewed as the way that is implemented the most widely in terms of maintaining the adaptive capacity of ports, while three measures (i.e., maintenance and resilience, crisis management, and business continuity plan) have similar popularity about the recovery capacity, as reflected by the occurrence probabilities in the model.



Fig. 4. Sub-network of 'Absorptive capacity'.



Fig. 6. Sub-network of 'Adaptive capacity'.

Adaptive\_capacity 🕅 69.54 Yes 30.46 No

## 5.2. Assessment and analysis of port resilience – scenario simulation

In this section, port resilience is assessed via the proposed OOBN model, and an analysis is conducted to clarify the variation of port resilience under dynamic situations.

As suggested by relevant studies (i.e., Hossain et al., 2019; Hossain et al., 2020) and opinions collected from experts, the values of parameters in assessing port resilience in Equation (3) should conform to the following principles:

The recovery degree  $\alpha$ , or the utilization rate, varies between 0.8 and 1.0; and

The recovery process in the second stage usually takes longer time than the performance reduction process in the first stage, indicating  $t_1 > t_2$ , or in other words, the ratio *rrs* ( $t_1/t_2$ ) varies in the interval (0,1].



Fig. 7. Sub-network of 'Port resilience'.



Fig. 8. The completed OOBN model for port resilience.





Fig. 9. The completed OOBN model for port resilience.

#### Z. Yang et al.

To demonstrate the assessment ability of the proposed OOBN model in this research, an example is illustrated in Fig. 9. The recovery degree  $\alpha$  is set as the median value 0.9, and the *rrs* is set as 1 ( $t_1 = t_2$ ), indicating the influenced time will not affect the final assessment result as the recovery time is equal to the lost time.

Once all information is put into the model, the panel on the left will present the calculated results. Under this scenario, it is found that port resilience is calculated as 0.82 under such conditions, along with the LP and RP calculated as 0.15 and 0.12, respectively. Port governments could follow this example to assess the port's resilience under their own conditions.

Further, as suggested by Equation (3), it is obvious that the port resilience is indeed affected by two parameters when the probabilities are determined by the OOBN model: the recovery degree (utilization rate)  $\alpha$  and the *rrs* (relative recovery speed)  $\beta$ . The clarification of the variation of the port resilience value along with the change of these two parameters plays a significant role in understanding the trend of port resilience dynamically.

*Matlab* is utilized to draw the surface diagram that is able to reveal the trend of port resilience along with the change of these two parameters, which is illustrated in Fig. 10. In addition, the results of multiple scenario simulations under dynamic situations are presented in Fig. 11, where the value of one parameter is changed while the value of another value is kept unchanged.

From Figs. 10 and 11, several new findings are obtained as follows:

When both parameter  $\alpha$  and  $\beta$  reach their maximum value '1', the port system at this point will have the strongest performance against the negative impacts caused by the disruptions, whose resilience value is around 0.907; while on the other hand, if the recovery time last for infinitely long when comparing with the time of the first stage, the port resilience will approach zero. This finding well reflects the reality and in part verifies the model and the results.

In general, whatever the defined parameter is, its value increment will lead to the improvement of port resilience. The simultaneous increase of two parameters will further improve the port resilience to a greater extent. This finding provides valuable insights for port governments to focus not only on the utilization rate of port facilities, but also pay enough attention on the relative recovery speed of adopted recovery measures or adaptive ways.

Additionally, for a specific point on the surface diagram, it is not difficult to find that the higher the value of  $\alpha$  and  $\beta$  it has, the larger the gradient it will have. Such phenomenon indicates the port resilience will be enhanced faster and faster along with the continuous increase of two parameters, indicating the faster *rrs* of adopted measures or higher utilization rate of port facilities should be the expected goals of port operations.

## 5.3. Sensitivity analysis

Sensitivity analysis is a widely applied measure to analyze the sensitivity of the resulting belief update of the target variable to the value variation of attribute variables. It can clarify the influence degree of different risk variables on the target variable and identify key variables that have stronger connections with the target variable, thus assisting the decision-makers in determining where they can improve, and how to formulate rational strategies.

As introduced in Section 3, mutual information analysis is applied in this research to figure out the influence magnitude of different variables. A basic principle of mutual information analysis is that a variable with a higher value of IMI is believed to have a greater influence on the target variable than one with a lower value. With the help of *Hugin* software, the results of IMI calculation in terms of different perspectives are presented in Table 3. Under each perspective, the influencing variables are listed in a descending order of value.



Fig. 10. Port resilience variation under different parameters.



a) Trend of port resilience along with the variation of b) Trend of port resilience along with the variation of  $\alpha$   $\beta$ 



In terms of the disruption perspective, Reduced navigational safety in port area is the risk type brought by climate change that has the greatest influence to cause disruptions of the port system in the GBA region, whose IMI value is way higher than other types. Disruption of maritime supply chains is at the second place, indicating the emerging issues of maritime supply chains due to the climate change may disrupt the port resilience to a certain extent. Following these two risk types, Damage to nearby ecosystems, Increased maintenance costs, Increased downtime and delays and Reduced cargo handling capacity have similar but less influential impacts on port disruptions, while power outages, increased risk of ship collision in port areas and damage to port infrastructure' are among top three risk factors that have the least impact on the occurrence of port disruption.

As to the absorptive capacity, it is found that advanced equipment has the largest impact, which is followed by the electronic exchange platform and the skillful emergency response team. These three measures have a similar value of mutual information, indicating the improvement or implementation of these aspects could increase the absorptive capacity of the port system in the GBA region effectively. On the other hand, the other two measures, the skilled labor with professional qualification and communication, have limited impact on absorptive capacity, demonstrating the implementation and promotion of these two measures are not effective, thus not worth investing more resources.

Among the ways to recover the performance of port climate resilience, maintenance and reliability is proved to exhibit the maximum effect on improving the recovery progress of the port system. Either preventive measures or corrective ways are useful, with the preventive actions can reduce the failure rate and the corrective actions can alleviate the potential consequences. Port innovation have similar influence, and it is not difficult to understand that the application of innovative technologies (e.g., digital tools, intelligent facilities) is of crucial importance in speed up the recovery progress. Port health and safety management, the management of risky operations that may harm the safety and health of cargo or staffs, also has a considerable effect on enhancing the recovery ability of a port system, although a little bit less than the former two measures. When it comes to the last two measures, the business continuity plan and crisis management policy, their influence could be ignored because of the extremely low value of mutual information.

Concerning the adaptation, technology restoration in the future is no doubt the strategy that plays a crucial role in helping the port system maintain at new status in the GBA region. Port worldwide network has the second highest value of mutual information with adaptive capacity, and then followed by the port community system and hinterland connection. In a word, these four strategies are viewed as the major ones in withstanding the disruptions caused by climate change and improving the resilience of the port system. In fact, during the interview stage, many practitioners proved that the port connections and the implementation of advanced technology have top priorities in dealing with climate change. Since climate change may affect the operations of port clusters within the jurisdiction of the GBA region, the enhancement in port connection and technology innovation could unite local ports and help resist potential hazards and disruptions. Service restoration also has a certain impact on the adaptive capacity, although not significant but non-ignorable. In addition, besides the negative effects brought by climate change, the appearance of such disruptions could also be an opportunity to promote the development of new technologies, i.e., data interoperability standardization, and cyber port infrastructure. However, because of the immaturity of such technologies, the influence on maintaining adaptation and improving port resilience is limited currently, as reflected by the value in Table 3. Shortly, it is expected that these digital strategies could play an increasingly important role in reducing potential risks, maintaining healthy port management, restricting the negative impacts caused by disruptions, and improving port resilience.

#### 5.4. Practical implications

Based on the analysis in the above sections, practical implications with regard to the enhancement of port resilience when facing climate-related risks can be put forward to help industry practitioner and port governments in port management.

(1) Formulation of specific instructions on controlling climate-related risks with significant impact on port disruptions.

When formulating regulations or rules, port governments should pay more attention on the identified climate-related risks with greater impacts on the occurrence of port disruptions, for example, the reduced navigational safety in port areas, the disruption of maritime supply chains, and the potential damage to nearby ecosystems. While for other risk types with less influence, i.e., increased maintenance costs and the increased downtime and delays, lower priorities should be assigned only when sufficient resources are possessed.

Specifically, when the navigational safety issue within port areas in the GBA region occurs, the occurrence probability of port disruptions will increase from 49.04 % to 59.72 %, as indicated by the proposed OOBN model. Similarly, this number will change to 57.33 % if the maritime supply chain is disrupted. Further, if both risks occur simultaneously, the occurrence probability will increase dramatically to 67.98 %. These changes indicate the likelihood of port systems being disrupted will be significantly increased under such conditions.

Therefore, port governments should formulate targeted instructions and implement countermeasures to control and reduce the likelihood of these identified climate-related risks. For example, it is promising to conduct the installation and improvement on monitoring system on vessel navigation safety within the port jurisdictions, and the development of digital early-warning system for the supervision of the real-time condition of maritime supply chain.

(2) Implementation of effective measures to enhance the port climate resilience across multiple aspects.

There are various measures applied to improve the port climate resilience across multiple perspectives, i.e., absorptive measures, recovery measures, and adaptive measures. However, through an in-depth analysis based on the proposed OOBN models in this research, it is found that not all of them are effective.

In terms of absorptive measures, the implementation of advanced equipment (e.g., automated equipment, facilities equipped with advanced technologies such as the internet of things, cyber infrastructure and the blockchain) in port management is found to be the most effective selection, which could improve the absorptive capacity from 69.37 % to 78.48 %, as estimated by the proposed OOBN model. Followed by the electronic exchange platform (e.g., single window, port community system), the application is proved to help transform the operation modes on port, leading to an increase in absorptive capacity probability to 77.48 %. In addition, a skillful emergency response team will also help absorb the potential climate-related disruptions to a certain extent, demonstrating the necessity of port authorities to build a team full of skillful and experienced personnel.

As to the recovery measures, maintenance and reliability of port systems is proved to be the most effective measure to help port system recover from potential disruptions, with a success rate of recovery capacity improved from 69.4 % to 80.24 %. Besides, the implementation of innovative technologies in port systems also have positive influence on speeding up the recovery progress, which is only slightly weaker than the maintenance and reliability (a success rate of 80.07 %). Meanwhile, port governments should keep an eye on the potential risky operations on ports and ensure the health and safety of cargo and staffs are guaranteed.

Eventually, to maintain a steady status, effective adaptive-related measures are suggested to be implemented, and the most recommended ones are the technology restoration and the port worldwide network. The technology restoration could help port government restore and reconstruct the port systems that may potentially affected by climate-related disruptions, as well as protect the port systems from suffering similar damage in the future again. As presented in the OOBN model, the implementation of the measure could effectively improve the adaptive capacity from 69.54 % to 79.13 %. In addition to this, port worldwide network is another effective adaptive measure. The development of port worldwide network involves the global container shipping network, port connectivity, port service, port supply chain, and other important components. The complicated network is proved to be effective in maintaining the port system at a steady status against possible climate change, as proved in the proposed model with an improvement from 69.54 % to 77.76 %. Other adaptive measures are far less effective on maintaining the steady status of port resilience performance than these two measures, for example, port community system, hinterland connection and service restoration are found to have limited influence, while the influence of data interoperability standardization and cyber port infrastructure is negligible.

In light of the above, port governments are advised to implement these identified effective measures from multiple perspectives to enhance the resilience performance of their port systems, which could make full use of possessed resources against possible climate-related disruptions and risks.

## 6. Conclusions

In the face of risks brought by climate change, ports are viewed as vulnerable infrastructures in dealing with potential disruptions from these extreme weather events, especially coastal ports in the GBA region, which are naturally vulnerable to such risks as sea level rise, typhoon and flooding. Understanding the resilience of ports when facing these risks is of essential importance to port management in the GBA region. Therefore, in this paper, a comprehensive resilience assessment framework is newly proposed via the development of a OOBN climate resilience model integrated with the EM learning approach, which could provide more reliable assessment results than traditional approaches. The influenced time of the disruption stage and the recovery stage, which are ignored in the previous research, is considered in the assessment of port resilience for the first time. Factors from multiple dimensions are selected, and a questionnaire is designed to collect relevant data and information about the GBA accordingly. Eventually, the obtained OOBN model containing five sub-networks is developed, which can assess the resilience of port systems dynamically.

Through an in-depth analysis of the developed model targeting the GBA, it is proved that the OOBN model could provide a comprehensible idea for coastal governments in evaluating the resilience and risk of port systems when facing climate change under various circumstances effectively and reliably. Moreover, it is evident that the port resilience to climate change will be enhanced with a higher utilization rate and a faster *rrs*. In addition, the influence of factors in different dimensions is clarified, especially the identification of those factors with significant impacts. Reduced navigational safety in port areas is identified as the risk factor that has the highest influence on the occurrence of port disruptions, while the advanced equipment, maintenance and reliability, and technology restoration in the future are viewed as the most effective measures in enhancing the resilience of port systems in terms of the absorptive capacity, recovery capacity, and adaptive capacity respectively.

Practically, these findings provide valuable insights and implications for port governments in developing effective port management plans and policies. It is advised that port governments should formulate targeted rules on identified high climate-related risks to control and reduce their occurrence likelihood, e.g., the development of digital real-time monitoring or early-warning systems on vessel navigation safety and maritime supply chain condition. Further, effective measures to help improve the resilience performance in different aspects are provided, which could assist the port authorities to make full use of possessed resources and enhance the port resilience to the maximum extent, i.e., the implementation of advance equipment and the electronic exchange platform in improving the absorptive capacity, the effort paid on port maintenance and innovative technologies in accelerating the recovery progress, and the application of the technology restoration and the port worldwide network in maintaining steady status of port performance when facing climate changes. Other measures should be placed at lower priorities only if there are extra resources available after the above prioritise measures being implemented.

In future research, our work will focus on the improvement of the applied methodology by introducing advanced data-driven machine-learning approaches, which could reduce the uncertainties caused by subjective judgment to a certain extent when developing the model structure. Besides, more areas will be included in the port resilience research, and a comparison analysis of port performance across different regions when facing climate change will be conducted, for example, the Southern East Asia (SEA) area to generate more empirical evidence in terms of rational of adaptation measures. This will help both academic researchers and industry practitioners better understand the resilience of port systems under different circumstances, thus promoting the proposal of rational port management strategies and regional collaborations among these closely linked areas.

## CRediT authorship contribution statement

Zhisen Yang: Writing – review & editing, Writing – original draft, Validation, Resources, Project administration, Methodology, Investigation, Funding acquisition, Formal analysis, Data curation, Conceptualization. Yui-yip Lau: Writing – original draft, Data curation. Mark Ching-Pong Poo: Writing – original draft, Formal analysis, Data curation. Jingbo Yin: Writing – original draft. Zaili Yang: Writing – review & editing, Validation, Resources, Project administration, Funding acquisition.

## Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

## Acknowledgments

This work is fully supported by grants from the National Natural Science Foundation of China (52301421), Humanities and Social Science Fund of Ministry of Education of China (23YJCZH274), Guangdong Office of Philosophy and Social Science (GD23XGL117), Research Grants Council of the Hong Kong Special Administrative Region China (UGC/FDS24/B07/22) and Natural Science Foundation of Top Talent of SZTU (GDRC202212). This research is also funded by a European Research Council project under the European Union's Horizon 2020 research and innovation programme (TRUST CoG 2019 864724).

The funding body will be acknowledged after peer review.

## Appendix A

Demographics	Number of Respondents	Percentage
Gender		
Male	64	64
Female	36	36
Age Group		
18-25 years	16	16
26-35 years	36	36
		(continued on next page)

 Table A1

 Description of Profile of Survey Respondents.

#### Table A1 (continued)

Demographics	Number of Respondents	Percentage
36-50 years	31	31
51-60 years	9	9
Above 60 years	8	8
Educational Level		
Secondary School Level	4	4
Diploma/Certificate	3	3
Associate Degree/Higher Diploma	6	6
Bachelor's degree	54	54
Master's degree	32	32
Doctoral Degree	3	3
Current Job Position		
Supervisor/Officer	28	28
Consultant	8	8
Assistant Manager	10	10
Manager	29	29
Director	11	11
Analyst	5	5
Others	11	11
Working Department		
Procurement	4	4
Warehousing	4	4
Transportation	37	37
Inventory	3	3
Customer Service	19	19
Research and Development	16	16
Others	19	19
Company's Nature		
Third-party logistics service providers	28	28
Freight forwarder	17	17
Transport operators	16	16
Port	1	1
Warehouse	6	6
Trading	14	14
Others	20	20
Working Experience		
2–5 years	34	34
6–9 years	19	19
10–15 years	12	12
16–20 years	11	11
21–25 years	12	12
26–30 years	7	7
31 years or above	7	7

## References

- Alamoush, A.S., Ballini, F., Ölçer, A.I., 2024. Investigating determinants of port decarbonisation implementation using the lens of Implementation Theory. Transp. Econ. Manage. 2, 90–111.
- Becker, A., Acciaro, M., Asariotis, R., Cabrera, E., Cretegny, L., Crist, P., Esteban, M., Mather, A., Messner, S., Naruse, S., Ng, A.K., Rahmstorf, S., Savonis, M., Song, D. W., Stenek, V., Velegrakis, A.F., 2013. A note on climate change adaptation for seaports: A challenge for global ports, a challenge for global society. Clim. Change 120 (4), 683–695.
- Becker, A., Ng, A.K., McEvoy, D., Mullett, J., 2018. Implications of climate change for shipping: Ports and supply chains. Wiley Interdiscip. Rev. Clim. Chang. 9 (2), e508.
- Calderón-Rivera, N., Bartusevičienė, I., Ballini, F., 2024. Barriers and solutions for sustainable development of inland waterway transport: A literature review. Transport Economics and Management 2, 31–44.
- Chen, Q., Zhang, H., Lau, Y.Y., Wang, T., Wang, W., Zhang, G., 2023. Climate change, carbon peaks, and carbon neutralisation: A bibliometric study from 2006 to 2023. Sustainability 15, 5723.
- Fan, S., Yang, Z., 2024. Accident data-driven human fatigue analysis in maritime transport using machine learning. Reliab. Eng. Syst. Saf. 241, 109675.

Fu, S.S., Zhang, Y., Zhang, M.Y., Han, B., Wu, Z.D., 2023. An object-oriented Bayesian network model for the quantitative risk assessment of navigational accidents in ice-covered Arctic waters. Reliab. Eng. Syst. Saf. 238, 109459.

Golan, M.S., Jernegan, L.H., Linkov, I., 2020. Trends and applications of resilience analytics in supply chain modeling: systematic literature review in the context of the COVID-19 pandemic. Environmental Systems and Decisions 40 (1), 222–243.

Gong, L., Xiao, Y.B., Jiang, C., Zheng, S., Fu, X., 2020. Seaport investments in capacity and natural disaster prevention. Transp. Res. Part D: Transp. Environ. 85, 102367.

Griese, K.-M., Franz, M., Busch, J.N., Isensee, C., 2021. Acceptance of climate adaptation measures for transport operations: Conceptual and empirical overview. Transp. Res. Part D: Transp. Environ. 101, 103068.

Gu, B., Liu, J., 2023. Port resilience analysis based on the HHM-FCM approach under COVID-19. Ocean Coast. Manag. 243, 106741.

Gu, B., Liu, J., Ye, X., Gong, Y., Chen, J., 2024. Data-driven approach for port resilience evaluation. Transportation Research Part e: Logistics and Transportation Review 186, 103570.

Guangdong Meteorological Service. (2024). Report of early warning big data in Guangdong Province in 2023.

Gürgen, S., Yazır, D., Konur, O., 2023. Fuzzy fault tree analysis for loss of ship steering ability. Ocean Eng. 279, 114419.

Henry, D., Ramirez-Marquez, J.E., 2012. Generic metrics and quantitative approaches for system resilience as a function of time. Reliab. Eng. Syst. Saf. 99, 114–122. Hossain, N.U.I., Nur, F., Hosseini, S., Jaradat, R., Marufuzzaman, M., Puryear, S.M., 2019. A Bayesian network based approach for modeling and assessing resilience: A case study of a full service deep water port. Reliab. Eng. Syst. Saf. 189, 378–396.

- Hosseini, S., Barker, K., 2016. Modeling infrastructure resilience using Bayesian networks: A case study of inland waterway ports. Comput. Ind. Eng. 93, 252–266. Hosseini, S., Ivanov, D., Dolgui, A., 2019. Review of quantitative methods for supply chain resilience analysis. Transportation Research Part e: Logistics and Transportation Review 125, 285–307.
- Hossain, N.U.I., Amrani, S.E., Jaradat, R., Marufuzzaman, M., Buchanan, R., Rinaudo, C., Hamilton, M., 2020. Modeling and assessing interdependencies between critical infrastructures using Bayesian network: A case study of inland waterway port and surrounding supply chain network. Reliability Engineering & System Safety 198, 106898.

Iacobucci, D., Churchill, G.A., 2010. Marketing Research: Methodological Foundations, 10th edition. Thomson South-Western, London.

- Izaguirre, C., Losada, I.J., Camus, P., Vigh, J.L., Stenek, V., 2021. Climate change risk to global port operations. Nat. Clim. Chang. 11 (1), 14-20.
- Jensen, F.V., Nielsen, T.D., 2007. Bayesian Networks and Decision Graphs. Information Science and Statistics. Springer, New York, NY, pp. 51-108.
- Jiang, C., Zheng, S., Ng, A.K., Ge, Y.E., Fu, X., 2020. The climate change strategies of seaports: Mitigation vs. adaptation. Transp. Res. Part D: Transp. Environ. 89, 102603.
- John, A., Yang, Z., Riahi, R., Wang, J., 2014. Application of a collaborative modelling and strategic fuzzy decision support system for selecting appropriate resilience strategies for seaport operations. Journal of Traffic and Transportation Engineering (english Edition) 1 (3), 159–179.
- John, A., Yang, Z., Riahi, R., Wang, J., 2016. A risk assessment approach to improve the resilience of a seaport system using Bayesian networks. Ocean Eng. 111, 136–147.
- Komugabe-Dixson, A.F., de Ville, N.S., Trundle, A., McEvoy, D., 2019. Environmental change, urbanisation, and socio-ecological resilience in the Pacific: Community narratives from Port Vila. Vanuatu. ecosystem Services 39, 100973.
- Kuzu, A.C., Akyuz, E., Arslan, O., 2019. Application of Fuzzy Fault Tree Analysis (FFTA) to maritime industry: A risk analysing of ship mooring operation. Ocean Eng. 179, 128–134.
- Lau, Y.Y., Chen, Q., Poo, M.C.P., Ng, A.K.Y., Chung, C.Y., 2024. Maritime transport resilience: A systematic literature review on the current state of the art, research agenda and future research directions. Ocean Coast. Manag. 251, 107086.
- Lauritzen, S.L., 1995. The EM algorithm for graphical association models with missing data. Comput. Stat. Data Anal. 19, 191-201.
- Li, D., Jiao, J., Wang, S., Zhou, G., 2024a. Energy import resilience evaluation considering maritime transport disruptions: A case of China. Transp. Res. Part D: Transp. Environ. 133, 104242.
- Li, H., Çelik, C., Bashir, M., Zou, L., Yang, Z., 2024b. Incorporation of a global perspective into data-driven analysis of maritime collision accident risk. Reliab. Eng. Syst. Saf. 249, 110187.
- Liu, J., Qi, Y., Lyu, W., 2023b. Port resilience in the post-COVID-19 era. Ocean Coast. Manag. 238, 106565.
- Liu, Q., Pérès, F., Tchangani, A., 2016. Object Oriented Bayesian Network for complex system risk assessment. IFAC-PapersOnLine 49 (28), 31-36.
- Liu, Q., Yang, Y., Ng, A.K.Y., Jiang, C., 2023c. An analysis on the resilience of the European port network. Transp. Res. A Policy Pract. 175, 103778.
- Maritime and Port Authority of Singapore. (2022). Business continuity plan. https://www.mpa.gov.sg/web/portal/home/port-of-singapore/business-continuity-plan. Maritime and Port Authority of Singapore. (2022). Singapore's container throughput. https://www.mpa.gov.sg/web/portal/home/maritime-singapore/port-of-singapore/singapore-container-throughput.

Madhusudan, C., Gangathy, G.P., 2011. Disaster resilience of transportation infrastructure and ports–An overview. International Journal of Geomatics and Geosciences 2 (2), 443–455.

- Mansouri, M., Nilchiani, R., Mostashari, A., 2010. A policy making framework for resilient port infrastructure systems. Mar. Policy 34 (6), 1125–1134.
- Masodzadeh, P.G., Ölçer, A.I., Ballini, F., Dalaklis, D., 2024. The contribution of ports to shipping decarbonisation: An analysis of port incentive programmes and the executive role of port state control. Transport Economics and Management 2, 191–202.
- Morris, L.L., 2020. Stakeholder collaboration as a pathway to climate adaptation at coastal ports. Marit. Policy Manag. 47 (7), 953-967.
- Ngai, E.W.T., Lai, K.H., Cheng, T.C.E., 2008. Logistics information systems: The Hong Kong experience. International Journal of Production Economics 113, 223–234.
- Ng, A.K., Wang, T., Yang, Z., Li, K.X., Jiang, C., 2018. How is business adapting to climate change impacts appropriately? Insight from the commercial port sector. J. Bus. Ethics 150, 1029–1047.
- Notteboom, T., Pallis, T., Rodrigue, J.P., 2021. Disruptions and resilience in global container shipping and ports: the COVID-19 pandemic versus the 2008–2009 financial crisis. Maritime Economics & Logistics 23, 179–210.
- Panahi, R., Ng, A.K.Y., Afenyo, M., Lau, Y.Y., 2021. Reflecting on forty years contextual evolution of Arctic port research: The past and now. Transp. Res. A Policy and Practice 144, 189–203.
- Panahi, R., Gargari, N.S., Lau, Y.Y., Ng, A.K.Y., 2022. Developing a resilience assessment model for critical infrastructures: The case of port in tackling the impacts posed by the COVID-19 pandemic. Ocean Coast. Manag. 226, 106240.
- Patel, M.M., Saltzman, L.Y., Ferreira, R.J., Lesen, A.E., 2018. Resilience: Examining the impacts of the Deepwater Horizon oil spill on the Gulf Coast Vietnamese American community. Soc. Sci. 7 (10), 203.
- Poo, M.C.P., Yang, Z., Dimitriu, D., Qu, Z., Jin, Z., Feng, X., 2021. Climate change risk indicators (CCRI) for seaports in the United Kingdom. Ocean & Coastal Management 205, 105580.
- Poo, M.C.P., Yang, Z., 2024. Optimising the resilience of shipping networks to climate vulnerability. Marit. Policy Manag. 51 (1), 15–34.
- Port of Hong Kong. (2024). Container throughput. https://www.mardep.gov.hk/en/publication/statistics.html.
- Port of Los Angeles. (2021). Sustainability. https://www.portoflosangeles.org/environment/sustainability.
- Port of Rotterdam. (2022). Sustainability. https://www.portofrotterdam.com/en/our-port/sustainability.
- Qin, X., Wang, S., Meng, M., 2024. SEA for better climate adaptation in the face of the flood risk: Multi-scenario, strategic forecasting, nature-based solutions. Environ. Impact Assess. Rev. 106, 107495.
- Rehak, D., Senovsky, P., Slivkova, S., 2018. Resilience of critical infrastructure elements and its main factors. Systems 6, 21.
- Rinaldi, S.M., Peerenboom, J.P., Kelly, T.K., 2001. Identifying, understanding, and analysing critical infrastructure interdependencies. IEEE Control Syst. Mag. 21 (6), 11–25.
- Roukounis, C.N., Tsihrintzis, V.A., 2024. Climate Change Adaptation Strategies for Coastal Resilience: A Stakeholder Surveys. Water 16 (11), 1519.
- Senol, Y.E., Aydogdu, Y.V., Sahin, B., Kilic, I., 2015. Fault Tree Analysis of chemical cargo contamination by using fuzzy approach. Expert Syst. Appl. 42 (12), 5232–5244.

Port of Shenzhen. (2024). Climate change adaptation. https://www.szport.gov.cn/english/climate-change-adaptation.html.

Sohu. (2018). https://www.sohu.com/a/254419257\_265147.

The Intergovernmental Panel on Climate Change. (2019). IPCC Special Report on Climate Change and Land, https://www.ipcc.ch/srccl/.

Verschuur, J., Koks, E.E., Hall, J.W., 2020. Port disruptions due to natural disasters: Insights into port and logistics resilience. Transportation Research Part d: Transport and Environment 85, 102393.

Wan, C., Yang, Z., Zhang, D., Yan, X., Fan, S., 2018. Resilience in transportation systems: A systematic review and future directions. Transp. Rev. 38 (4), 479–498.
 Wan, C., Yuan, J., Cao, D., Wang, T., Ng, A.K.Y., 2024. A fuzzy evidential reasoning-based model for evaluating resilience of ports to typhoons. Transp. Res. Part D: Transp. Environ. 133, 104228.

- Wang, L., Lau, Y.Y., Su, H., Zhu, Y., Kanrak, M., 2022. Dynamics of the Asian shipping network in adjacent ports: Comparative case studies of Shanghai-Ningbo and Hong Kong- Shenzhen. Ocean Coast. Manag. 221, 106127.
- Wang, L., Tang, Z., Lau, Y.Y., Zhou, J., Kanrak, M., 2024a. The promotion of Shanghai's position and function in the global container shipping network. Int. J. Urban Sci. 1–21.

Wang, N., Wu, M., Yuen, K.F., 2023a. Assessment of port resilience using Bayesian network: A study of strategies to enhance readiness and response capacities. Reliab. Eng. Syst. Saf. 237, 109394.

Wang, T., Poo, M.C.P., Ng, A.K.Y., Yang, Z., 2023b. Adapting to the Impacts Posed by Climate Change: Applying the Climate Change Risk Indicator (CCRI) Framework in a Multi-Modal Transport System. Sustainability 15 (10), 8190.

Wang, T., Ng, A.K.Y., Wang, J., Chen, Q., Pang, J., Tang, J., 2024b. Adaptation planning of container ports in the context of typhoon risks: The case of Ningbo-Zhoushan port in China. Ocean Coast. Manag. 257, 107303.

Wu, X., Wang, K., Fu, X., Jiang, C., Zheng, S., 2024. How would co-opetition with dry ports affect seaports' adaptation to disasters? Transp. Res. Part D: Transp. Environ. 130, 104194.

Wang, Y., Liu, C., Wang, Y., Liu, Y., Liu, T., 2024c. Climate risk assessment and adaption ability in China's coastal urban agglomerations - A case study of Guangdong-Hong Kong-Macao greater bay area. J. Clean. Prod. 451, 142036.

Xia, W., Mishra, J., Adulyasak, Y., 2024. Seaport adaptation and capacity investments under inter-port competition and climate-change uncertainty. Transp. Res. Part D: Transp. Environ. 130, 104183.
 Xu, L., Huang, J., Cehn, J., 2023. How does the initiative of 21st century maritime silk road incentive logistics development in China's coastal region? Ocean & Coastal

Management 239, 106066.

Xu, M., Zhu, Y., Liu, K., Ng, A.K.Y., 2024. Assessing resilience of global liner shipping network to tropical cyclones. Transportation Research Part d: Transport and Environment 131, 104189.

Yang, Z., Lun, V., Lagoudis, I.N., Lee, P.T.W., 2018a. Container transportation: resilience and sustainability. Transportation Research Part d: Transport and Environment 61, 420–422.

Yang, Z.S., Yang, Z.L., Yin, J.B., 2018b. Realising advanced risk-based port state control inspection using data-driven Bayesian networks. Transp. Res. A Policy Pract. 110, 38–56.

Yang, Z.S., Lau, Y.Y., Kanrak, M., 2023a. Pollution prevention of vessels in the greater bay area: A practical contribution of port state control inspection system towards carbon neutralisation using a tree augmented navies bayes approach. J. Clean. Prod. 423, 138651.

Yang, Z.S., Wan, C.P., Yang, Z.L., Yin, J.B., Yu, Q., 2023b. A machine learning-based Bayesian model for predicting the duration of ship detention in PSC inspection. Transportation Research Part e: Logistics and Transportation Review 180, 103331.

Yang, Z.S., Yu, Q., Yang, Z.L., Wan, C.P., 2024a. A data-driven Bayesian model for evaluating the duration of detention of ships in PSC inspections. Transportation Research Part e: Logistics and Transportation Review 181, 103371.

Yang, Z.S., Lau, Y.Y., Lei, Z.M., 2024b. Analysis of pollution prevention performance of vessels in Southeast Asia: Implications towards vessel emission control and reduction. Ocean Coast. Manag. 248, 106942.

Zhao, C., Yin, Y., Dong, K., Yang, C., Wang, K., 2024. How does port alliance survive extreme weather? Analysis using stochastic evolutionary game. Transp. Res. Part D: Transp. Environ. 133, 104294.

Zheng, K., Xian, Z., Liao, W., Chen, Y., 2024a. Urbanization impacts on sequential flood-heatwave events in the Guangdong-Hong Kong-Macao Greater Bay Area. China. Urban Climate 55, 101878.

Zheng, S., Jiang, C., Wang, K., 2024b. Climate change adaptation projects of transport facilities: is a demonstrator necessary? Annals of Operations Research. https:// doi.org/10.1007/s10479-024-06010-8.

Zhou, C., Xu, J., Miller-Hooks, E., Zhou, W., Chen, C.-H., Lee, L.H., Chew, E.P., Li, H., 2021. Analytics with digital-twinning: A decision support system for maintaining a resilient port. Decis. Support Syst. 143, 113496.

Zhou, K., Xing, W., Wang, J., Li, H., Yang, Z., 2024. A data-driven risk model for maritime casualty analysis: A global perspective. Reliab. Eng. Syst. Saf. 244, 109925.